Dear Editors and Reviewers:

Thank you for your letter and the reviewers’ comments concerning our manuscript (NO: 73193). We quite appreciate the reviewer’s insightful comments that drove us to carefully check and refine our manuscript. Our revisions and point-by-point responses to the referee’s questions are included below. We sincerely hope that the current version of this manuscript will be finally accepted for publication in *World Journal of Clinical Cases*.

**Reviewers’ comments:**

Reviewer 1

Q: this case is no novel. adrenal TB is common in developing nation. work up is straightforward.

Reply: Actually, adrenal TB is not that rare in developing countries, but in the present case, only the right adrenal gland was infected by TB. Isolated adrenal TB might be a rare clinical entity. Moreover, the CT image characters of the TB case was unique.

Reviewer 2

Q: The article reports a rare case of tuberculosis affecting unilateral adrenal. Well presented and discussed case. I suggest it be accepted for publication. Writing can be better with small adjustments, for better flow of text.

Reply: We have sought an English editing service from American Journal Expert to prepare the revised manuscript and we hope the revised one may meet the requirement.

**EDITORIAL OFFICE’S COMMENTS**

(1) Science editor:

The authors report a case of unilateral tuberculosis of the adrenal gland. The manuscript is well-written and concise. Adrenal involvement in disseminated tuberculosis is not uncommon, particularly in endemic regions of the globe, but this report can increase awareness of this atypical (unilateral) presentation in regions with
a smaller number of patients suffering with tuberculosis. Most of the references are from more than 10 years ago, adding more recent references could increase the value of the discussion section.

Language Quality: Grade B (Minor language polishing)

Scientific Quality: Grade C (Good)

Reply: We have added several recent references in the revised manuscript. There were 27 references in the revised manuscript and 19 of those were within recent 10 years.

(2) Company editor-in-chief:

I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. However, the quality of the English language of the manuscript does not meet the requirements of the journal. Before final acceptance, the author(s) must provide the English Language Certificate issued by a professional English language editing company. Please visit the following website for the professional English language editing companies we recommend: https://www.wjgnet.com/bpg/gerinfo/240. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

Reply: We have sought premium editing help from American Journal Expert to prepare the revised manuscript and will provide an English Language Certificate. The figures have been arranged by PowerPoint.